# Journal of Radiotherapy in Practice

cambridge.org/jrp

## **Literature Review**

**Cite this article:** Davies LSC and Drury-Smith H. (2023) The psychosocial responses of patients in cancer clinical trials: are they a barrier to participation? *Journal of Radiotherapy in Practice.* **22**(e44), 1–6. doi: 10.1017/ S1460396922000139

Received: 19 December 2021 Revised: 30 March 2022 Accepted: 1 April 2022

#### Key words:

cancer clinical trials; cancer patient perspectives; psychosocial barriers to participation

#### Author for correspondence:

Lucy Siew Chen Davies, The Christie NHS Foundation Trust, Manchester, UK. E-mail: l.davies33@nhs.net

© The Author(s), 2022. Published by Cambridge University Press.



# The psychosocial responses of patients in cancer clinical trials: are they a barrier to participation?

## Lucy Siew Chen Davies<sup>1</sup> and Heather Drury-Smith<sup>2</sup>

<sup>1</sup>Radiotherapy, The Christie NHS Foundation Trust, Manchester, UK and <sup>2</sup>Radiotherapy and Oncology, Allied Health Department, Sheffield Hallam University, Sheffield, UK

#### Abstract

*Introduction:* This paper aims to discuss the psychosocial concomitants with involvement in oncology clinical trials, focusing on barriers that can impact upon participation. It will conclude with some recommendations for strategies to address potential psychosocial barriers with the aim of increasing trial participation rates.

*Materials and methods:* A literature search was carried out using CINAHL, PubMed and EMCare databases with the following keywords for filtering: psychological distress, clinical trials, participation and oncology. The final selection of papers that met the inclusion criteria for this review was manually subjected to Critical Appraisal Skills Programme tool for relevance. *Results:* Thirteen papers were included in the review. The dominant theme within the literature is psychosocial obstacles to oncology clinical trial participation. Five key barriers were identified: anxiety and fear; ethnicity and social background; tensions between scientific objectives and personal motivations to participation; tensions between personal benefits versus altruism; carer perspectives.

*Conclusions:* The key barriers discussed led to the identification of a set of strategies to help mediate conflicting tensions and motivations of trial enrolment with a view to increasing participation rates. Further prospective research garnering primary data investigating both the psychological and psychosocial factors influencing cancer clinical trial participation for patients needs to be undertaken.

### Introduction

Clinical trial results provide the crucial evidence base for establishing the efficacy and safety of novel therapeutic approaches for malignancy before such cancer therapies can be implemented into the clinical setting. Fundamental to cancer clinical research is patient participation. Clinical trials can be carried out more effectively with greater participation, resulting in new treatments being implemented sooner.<sup>1</sup> Moreover, clinical trials may give opportunities for participants to access the newest available therapies.

However, despite over a decade of research into potential barriers, participation to cancer clinical trials remains low. Although the UK, comparatively speaking, has higher rates of clinical trial participation, with 30% of patients reporting that cancer research had been discussed with them,<sup>2</sup> there is a global concern that fewer than 5% of adult patients enrol in clinical trials.<sup>3,4</sup> Poor accrual results in trials remaining open for longer, potentially increasing costs and reducing the opportunity to improve patient outcomes.<sup>5</sup>

The barriers of patient participation in cancer clinical trials have been the topic of frequent review, with findings identifying a multitude of concerns influencing the decision to either accept or decline participation.<sup>6-8</sup> More often, the literature has focused on key structural barriers such as cost and travel,<sup>1</sup> and the accessibility of trials suitable for an individual's cancer type and stage,<sup>9</sup> rather than potential psychological obstacles to participation. However, more recently there is an increasing body of evidence highlighting emotional distress and the psychological burden of a cancer diagnosis as a key inhibitor.<sup>10-12</sup> One problem is that patients are often asked to consider enrolling onto a trial at a significant psychological time point, for example, just after confirmed diagnosis. At such times, a cancer diagnosis combined with the implications of clinical trial participation can instigate significant psychological effects for both patients and their carer's.<sup>13</sup> Identifying and acknowledging the psychosocial needs of clinical trial participants can potentially be used as a means to increase and maintain participation. This discussion paper will review the psychological experiences and psychosocial barriers to patients participating in cancer clinical trials, not just at the point of recruitment, but throughout the trial episode. This paper will also discuss possible strategies for addressing psychosocial barriers which in turn may enhance the participation rate of cancer clinical trials in the future.

#### **Methods**

Three key electronic databases were searched for literature related to psychosocial aspects of participation in cancer clinical trials: CINAHL; PubMed; EMCare. The following search strategy was used: (((((distress × OR anxiety OR depress × OR stress OR needs OR psychosocial OR "psycho social" OR psychological OR holistic).ti, ab OR ("PSYCHOLOGICAL DISTRESS"/ OR exp ANXIETY/ OR DEPRESSION/ OR "STRESS, PSYCHOLOGICAL"/)) AND (("clinical trial").ti,ab OR exp "CLINICAL TRIALS"/)) AND ((participation OR participating).ti,ab). Titles of papers from the search were scanned to identify articles pertinent to the central aim. Duplicates and papers not meeting the inclusion criteria outlined in Table 1 were removed. The relevance of potential papers was determined by further scrutiny of each paper's abstract with reference to the inclusion and exclusion criteria.

The literature search yielded few results of high-level evidence research papers (for example, randomised-controlled trials) despite entering different search terms and key words. The majority of outputs were review articles. The literature review was further supplemented with a manual search. This was carried out by scrutinising the reference list of relevant papers in relation to the search criteria, yielding further articles to ensure other applicable studies were included. Commentary articles and anecdotal accounts were also included to add to the discussion.

Thirteen articles were included for discussion in the main review. This body of literature was reviewed and appraised using the appropriate Critical Appraisal Skills Programme (CASP) quality assessment tool.<sup>14</sup> The characteristics and main findings of the studies were extracted and reported. Key themes related to the central topic of the psychosocial barriers to cancer clinical trial participation were identified: anxiety and fear; ethnicity and social background; tensions between scientific objectives and personal motivations to participation; tensions between personal benefits versus altruism; carer perspectives. These formed the framework for the current discussion.

# The Psychosocial Barriers to Cancer Clinical Trial Participation

#### *Psychological concomitants to a cancer diagnosis*

A cancer diagnosis can evoke a wide range of psychological responses, from natural reactions of vulnerability through to feelings of anxiety, fear, depression and spiritual crisis.<sup>15-17</sup> Furthermore, patients are commonly confronted by complex treatment options that are frequently imbued with uncertainty and mystification.<sup>18</sup> It is in this context that additional difficult decisions may be presented to patients if they are asked to consider the option to enrol in a cancer clinical trial.

#### Anxiety and fear as a barrier to participation

It is well established that anxiety and fear are frequently experienced by patients who have been diagnosed with cancer, sometimes referred to as 'cancer fear'.<sup>19</sup> Several authors have found fear applies not just to considering alternative treatment options, but to participation in clinical research.<sup>7,20</sup> A study by Quinn et al (2012), examined the role of fear in cancer patients' and their perceptions of clinical trial participation using semi-structured in-depth interviews.<sup>21</sup> Fear was experienced in several different contexts along the cancer pathway: the fear of the 'unknowns associated with cancer, cancer treatment, and clinical trials'.<sup>21</sup> Patients can lose a sense of control over their lives as they advance along the

Table 1. Inclusion and	l exclusion criteria
------------------------	----------------------

Inclusion	Exclusion
Publication dates of papers between 2000 and 2020	Publication dates of papers pre-2000
Articles written in the English language	Articles not written in the English language
Full text accessible	Unpublished literature
Psychological or psychosocial patient factors	Articles not addressing patient's psychological or psychosocial needs
Articles related to participation in oncology clinical trials	Clinical trials not within or related to oncology

cancer therapy trajectory and begin to navigate their way through a network of medical appointments and procedures which now affect their day-to-day lives. In this study, some patients reported that the reason they declined participation was a way of assuming some element of control over their own health-related decisions. Although this paper raises some interesting insights and draws valid conclusions to fear as a potential barrier to participation, the study sample was conveniently selected. Participants who were interviewed ranged in age from 32 to 75 years (mean age = 66) and although had a diagnosis of different cancer types, the sample consisted predominantly of lung cancer patients. The authors do not acknowledge or specify if the treatment intent was palliative or radical. Results, therefore, may not be generalisable to all cancer patient populations and other clinical trial scenarios.

A recent systematic review by Sheridan et al (2020) found fear to be the leading barrier to trial participation in a large number of studies.<sup>12</sup> Often this was associated with the perceived risks of treatment and associated potential toxicities. However, the review also highlighted altruism in terms of contributing to science and helping others was a commonly reported factor.

Schaefer et al (2001) also found fear as a central theme in their study.<sup>22</sup> They evaluated the decision-making process by women considering participation in a breast cancer prevention trial. In this case, fear was not so much related to the cancer diagnosis, but associated with the uncertainty of the effects of treatment. However, Schaefer et al (2001) also identified some participants who, despite expressing fear, did not want others to experience the devastation of breast cancer and for this reason were, nevertheless, happy to participate in the trial.<sup>22</sup> So for some patients, fear was ultimately not a barrier to enrolment. There were other women in the study who expressed fear related to distrust towards the medical profession. This mistrust appeared to arise as a result of the process of discussing the extent to which they met the eligibility criteria to participate in the trial, which resulted in a lack of trustworthiness. There was a perceived arbitrariness about eligibility. The issue of trust between potential participants and healthcare professionals (HCPs) is discussed further below. In evaluating the results of Schaefer et al.'s (2001) study, it should be noted that there are some limitations. Firstly, the participants included in the study do not have a confirmed breast cancer diagnosis and findings cannot be directly compared to those that have been diagnosed with cancer. Secondly, all respondents were Caucasian women with a relatively high proportion educated to Masters Level, which may not reflect the true cancer population. Nevertheless, the paper was included in this review as it provides a rich holistic and valuable qualitative insight of the process of decision-making.

#### Barriers related to ethnicity and social background

Another barrier to trial participation that has been found to be particularly relevant in relation to patients from black and minority ethnic groups is the fear and mistrust of HCPs. It has been noted that there is an underrepresentation of patients from ethnic minorities in cancer clinical trials.<sup>23</sup> In a review specific to minorities' participation in cancer trials, Salman et al. (2016) found fear and mistrust over the patient relationship with the HCP to be a key issue.<sup>10</sup> There were linguistic and literacy barriers giving rise to ineffective communication between patient and HCP. They also reported that cultural values, beliefs and stigma were a further concern. This was particularly highlighted by conflicting cultural beliefs in relation to herbal medicine or other complimentary therapies which may be contradictory with the scientific objectives of a clinical trial. This was found to lead to a resistance to participate in research.

Another social factor acting as a barrier to cancer clinical trial participation is the underrepresentation of poorer socio-economic groups, who appear to have more limited opportunities to participate.<sup>24,25</sup> Researchers seeking participants find such groups harder to reach. Conversely, patients from higher socio-economic groups have been found to be much more likely to enrol in cancer clinical trials.<sup>26,27</sup> An internet-based survey of 5499 patients, who were asked about their cancer treatment decisions, ascertained that income played a key role in the association of clinical trial participation.<sup>28</sup> Results from the survey indicate that discussion and participation of a trial were correlated with the participants of the younger, affluent and more highly educated population. In order for the results of clinical trials to be generalisable, the study sample and data collected needs to be representative of the general cancer population.

Ethnicity and socio-economic status can also intersect. Collectively, these factors can result in a lack of diversity within study samples and therefore potentially limit the generalisability of results to wider populations.<sup>26</sup> In summary, the research suggests that there are potentially inherent biases within participant selection in many cancer clinical trials, in particular in respect of ethnicity and socio-economic status. A related issue in oncology research undertaken in America is that potential participants might decide not to take part if they are unsure what the medical insurance implications of participation.<sup>25</sup>

# Barriers related to randomisation: scientific objectives versus personal hopes

The methods used for scientific objectivity in a clinical trial may conflict with individual participant's personal motivations to take part. This is particularly evident in relation to the need for randomisation.<sup>29</sup> A common perception reported in the literature is misconceptions of being the subject of experimental testing, with some patients comparing clinical trial participation to being treated 'like a guinea pig<sup>,5</sup> A large survey by Jenkins et al (2010) researched the attitudes of 1066 patients and their likely participation in a hypothetical two-arm randomised-controlled trial (RCT).<sup>30</sup> The concept of random assignment and leaving treatment decisions to chance was reported as a potential significant barrier to trial participation for patients. The results displayed a high proportion of patients (91%) were supportive of the notion that patients should be asked to take part in medical research. However, if treatment allocation was decided upon by randomisation, this figure dropped to 55%.

Harrop et al (2016) investigated the motivations, understandings and experiences of patients with advanced lung cancer who were participating in a non-placebo Clinical Trial of an Investigational Medicinal Product.<sup>31</sup> Ten participants from both the control and intervention arm of the FRAGMATIC trial were recruited to the integrated qualitative study (QualFRAG).<sup>32</sup> Participants were interviewed at up to three time points during their time in the trial to explore the psychological impact of participation in a clinical trial. Although it was reported that a generally positive view of participation in the FRAGMATIC trial was viewed, one of the interesting findings concerning randomisation, was that participants' understanding of the randomisation process was generally mixed, with many poorly comprehending the concept. Some patients understood and accepted the outcome of the randomisation. However, the study reported an extreme case where one respondent, who was randomised to the control group, was so 'disappointed and angered' by the result that they no longer wanted to continue taking part in the trial and refused further interviews. These negative emotional responses of allocation to the control arm ultimately lead to the individual's attrition from the trial. Although the authors were able to explore participants' concerns of randomisation in-depth, providing valid conclusions as to patients' understanding of random treatment allocation, some limitations to the findings should be highlighted. The participants were only interviewed once they had been allocated a treatment arm and therefore the study did not capture the potential shift of attitudes before and after treatment allocation. Furthermore, the study sample consisted of only ten participants, nine of which were male. Thus, conclusions cannot be drawn about other populations.

It is conceivable that the complex nature of RCT study design can contribute to patient misunderstanding and possibly deter patients from enrolling into cancer clinical trials. From the patients' perspective, in the context of life-threatening illness, it can seem a strange way to determine treatment.<sup>32</sup>

#### Hope: individual treatment benefit versus altruism

Although the literature highlights fear and anxiety as a barrier to participation to cancer research, it is worth addressing patients may also experience positive psychological outcomes of participation. Indeed, some authors have identified a key motivator for patients' enrolling in cancer research is 'hope'.<sup>13,33</sup> Chi (2007) considers hope to be an effective coping strategy, playing a significant role in aiding the cancer patient to adapt to their disease and treatment.<sup>34</sup> The notion of hope can change over time whereby people can transition from fear to hope.

One of the main themes in Dellson et al's (2018) interview study, which focused on patients with incurable cancer and their reasons to participate in cancer research, was the hope that they could overcome their cancer.<sup>35</sup> Some of the patients in this paper also reported an altruistic motivation whereby hope of recovery was not only in relation to themselves but for patients who may benefit from their participation in the future. In this respect, hope and altruism are intertwined. Similar findings were reported in a questionnaire-based study by Jenkins et al (2013), although in this case, the primary motivator for enrolment was altruism.<sup>32</sup> This contrasts with the results from a large cross-sectional survey by Truong et al (2011) where individuals identified the 'single most important' reason for choosing to participate in trials was to receive medical benefits and less likely for altruistic reasons.<sup>36</sup> For some patients, for example, individuals with a poorer prognosis or those entering earlier phase trials, altruism plays a less prominent role as a motivator for participation compared to expectations of personal benefit.33,37

If taking part in an oncology trial has the effect of raising individuals hope for themselves, then that may imply a potential ethical concern for HCPs in how they promote the potential benefits of trial for patients. There needs to be an awareness of the possibility of raising unrealistic hopes. An additional related concern is that in raising hopes in patients, this potentially could undermine their autonomy in their decision-making as to whether to participate. It becomes critically important that informed consent is based on the realities of trial participation.

#### Carer perspectives

When exploring the potential psychosocial obstacles of recruitment into cancer research, it is important to also consider the perceptions of the patient's family or caregiver. Kim and Given (2008) acknowledge that cancer, and its consequential impact on quality of life, is increasingly recognised as a family's concern as opposed to solely the patient's with the diagnosis.<sup>38</sup> The authors determine family members and friends may be a source in providing 'informal cancer care', which can often involve meeting the individual's multi-dimensional requirements. This can include symptom management, psychological and psychosocial support, assistance with transport and possibly, financial support.<sup>39</sup> In such instances, the decision to enter a cancer clinical trial is rarely a decision made purely by the patient alone. In Bell and Balneaves' (2015) review, they identified several studies where recommendations from support networks (family and friends) as being highly associated with clinical trial participation.<sup>5</sup> Some research studies may require more attendances to the hospital, resulting in potential logistical challenges and requiring the support of family members. It is therefore important that the HCP promotes open discussion amongst patients and their caregivers when considering cancer trial entry and along the trial pathway.

#### Strategies and facilitators to improve participation

#### Healthcare professional communication and informed consent

The role and influence of the HCP can play a key part in facilitating intervention strategies to improve participation. A fundamental theme arising from much of the literature reviewed above is the importance of a trusting relationship between the patient and the HCP. This appears to be an overriding factor determining patients' decisions about cancer trial participation.<sup>5</sup> Albrecht et al (2008) found that the 'quality and quantity' of effective communication between the HCP and the patient influenced the patient's decision and decision-making process.<sup>40</sup> They particularly emphasised the importance of HCPs facilitating the building of a sense of alliance between patient and physician. The decisions made in cancer treatment, and indeed, the decision to enter a cancer trial is one that should be made autonomously by the patient. Understandably, the complex nature of study trial design can bring about challenges for patients and their families to actively engage in making such decisions. Albrecht et al.'s (2008) study identified that the patients who were noticeably engaged in interacting with their doctors were more likely to be engaged in clinical trial entry.<sup>40</sup> HCPs should therefore be aware of their influence in promoting patient-centred care, providing supportive communication and delivering medical content in a language that patients and their families from all backgrounds understand to enable the patient to feel empowered to make additional decisions about trial participation.

Dellson et al. (2018) suggested the decision to participate in cancer research is instantaneous and guided by emotion,

significantly when based on a trusting relationship with the HCP.<sup>35</sup> In the light of the possibility of impulsive decision-making, it is crucial that full informed consent is nevertheless provided. Furthermore, care needs to be taken not to blur therapeutic and research ethics boundaries especially when securing informed consent. Researchers have a commitment to adhere to the principles of *Good Clinical Practice*, the international ethical, scientific and practical standard to which all clinical research is conducted to, as set out by the Health Research Authority (HRA) and underpinned in UK policy.<sup>41</sup> Reflective practice when applied to research is conceived essential to help manage those biases so it does not unduly influence the research outcomes.<sup>42</sup>

#### Psychological and psychosocial interventions

The objectivity of a cancer clinical trial is to test the efficacy and safety of a new therapeutic approach. Historically, cancer care and clinical trials placed more importance on physical toxicities and clinical outcomes over the psychological and psychosocial issues.<sup>43</sup> However, this approach has since shifted, and the importance of psychosocial care for cancer patients has become more critical, with emphasis highlighted in key clinical guidance where recommendations for routine screening to measure psychological distress should be carried out.<sup>44</sup> HCPs are in a key position to identify barriers and signpost to appropriate services.

Measuring toxicity utilises standardised nationally accepted systematic methods for reporting in cancer clinical trials.<sup>45</sup> Distress can be considered a psychological toxicity. Although there are some available standardised screening tools for assessing it, such as the National Comprehensive Cancer Network's *Distress Thermometer*,<sup>46</sup> they retain an element of subjectivity and therefore HCPs in this area of care may feel less confident in evaluating the significance of the outcomes. The National Institute for Clinical Excellence (NICE) has provided some guidance for HCPs, such as the *Model of Psychological Intervention*.<sup>44</sup>

#### Education and training

Those involved in clinical trials, particularly those leading clinical trials, need to be fully conversant with the factors that influence patient participation. There are specific areas of education and training which they and the research team may benefit from.<sup>5</sup> This in particular may include how to assess psychosocial concerns and distress and the application of standardised tools and evaluating the outcome of those assessments.

It is likely that if those involved in cancer clinical trials possess the qualities associated with an effective HCP, then they are also likely to be effective at addressing psychosocial concerns that potential participants might have. Advanced communication skills training for HCPs working in a research role within cancer care may be advantageous in providing HCPs with the communication and reflective practice skills to individualise their responses to patients' emotions and enable them to offer appropriate support when considering clinical trial entry.

#### **Conclusion and Recommendations**

Psychological responses to patients participating in cancer clinical trials can be a challenging issue to mediate. The research literature does not provide, at present, definitive clinical guidance. This is for several reasons: much of the research relates to different disease sites which may have differing prognoses; not all of the studies address the various phases of clinical trials. Furthermore, consideration has to be taken into account for various methodological limitations within studies undertaken to date. However, the research that has been undertaken does provide valuable insights.

The following key messages are identified. Firstly, there are many psychological and psychosocial factors impacting on cancer clinical trial participation. However, the psychosocial responses are not universally negative and for some individuals, a positive emotional response can act as a motivator to participation. Secondly, HCPs have a fundamental role in overcoming potential obstacles to participation in developing positive relationships with their patients. Thirdly, psychological assessments, although now more likely to be undertaken, could benefit from a more standardised and expert approach. There is a dearth of literature on existing established interventions in place to support patients and their decision to enrol in a cancer clinical trial. Further systematic research is warranted that considers these factors when developing effective interventions and strategies to greater understand patients' psychological and psychosocial needs to improve cancer clinical trial participation rates.

#### Acknowledgements. None.

Financial support. Self-funded, no financial support.

Conflict of interest. The authors declare none.

#### References

- Unger J, Vaidya E, Hersnman DL, Minasian LM, Fleury ME. Systematic review and meta-analysis of the magnitude of structural, clinical, and physician and patient barriers to cancer clinical trial participation. J Natl Cancer Inst 2019; 111 (3): 245–255.
- NHS England. National Cancer Patient Experience Survey 2019 National Results Summary. 2020. https://www.ncpes.co.uk/wp-content/uploads/ 2020/06/CPES-2019-National-Report\_V1.pdf. Accessed on 16<sup>th</sup> September 2020.
- Fouad MN, Lee JY, Catalano PJ et al. Enrolment of patients with lung and colorectal cancers onto clinical trials. J Oncol Pract 2013; 9 (2): e40–e47.
- Unger J, Cook E, Tai E, Bleyer A. Role of clinical trial participation in cancer research: barriers, evidence, and strategies. Am Soc Clin Oncol Educ Book 2016; 35: 185–198.
- Bell JA, Balneaves LG. Cancer patient decision making related to clinical trial participation: an integrative review with implications for patients' relational autonomy. Support Care Cancer 2015; 23: 1169–1196.
- Biedrzycki BA. Decision making for cancer clinical trial participation: a systematic review. Oncol Nurs Forum 2010; 37 (6): E387–E399.
- Mills EJ, Seely D, Rachlis B et al. Barriers to participation in clinical trials of cancer: a meta-analysis and systematic review of patient-reported factors. Lancet Oncol 2016; 7 (2): 141–148.
- Townsley CA, Selby R, Siu LL. Systematic review of barriers to the recruitment of older patients with cancer onto clinical trials. J Clin Oncol 2005; 23: 3112–3124.
- Vardy J, Dadasovich R, Beale P, Boyer M, Clarke SJ. Eligibility of patients with advanced non-small cell lung cancer for phase III chemotherapy trials. BMC Cancer 2019; 9: 130.
- Salman A, Nguyen C, Lee Y, Cooksey-James T. A review of barriers to minorities' participation in cancer clinical trials: implications for future cancer research. J Immigr Minor Health 2016; 18: 447–453.
- Kao CY, Aranda S, Krishnasamy M, Hamilton B. Interventions to improve patient understanding of cancer clinical trial participation: a systematic review. Eur J Cancer Care 2017; 26 (2): e12424.
- Sheridan R, Martin-Kerry J, Hudson J, Parker A, Bower P, Knapp P. Why do patients take part in research? An overview of systematic reviews of psychosocial barriers and facilitators. Trials 2020; 21: 259.

- Kelly C, Ghazi F, Caldwell K. Psychological distress of cancer and clinical trial participation: a review of the literature. Eur J Cancer Care 2002; 11 (1): 6–15.
- CASP. Critical Appraisal Skills Programme (CASP) 2021. https://casp-uk. net/casp-tools-checklists/. Accessed on 7<sup>th</sup> March 2021.
- Holland JC. History of psychosocial oncology: overcoming attitudinal and conceptual barriers. Psychosom Med 2002; 64 (2): 206–221.
- Schumacher JR, Palta M, LoConte NK et al. Characterizing the psychological distress response before and after a cancer diagnosis. J Behav Med 2013; 36: 591–600.
- Tondorf T, Grossert A, Rothschild SI et al. Focusing on cancer patients' intentions to use psychooncological support: a longitudinal, mixed- methods study. Psycho-Oncol 2018; 27 (6): 1656–1663.
- Reyna VF, Nelson WL, Han PK, Pignone MP. Decision making and cancer. Am Psychol 2015; 70 (2): 105–118.
- Vrinten C, McGregor LM, Heinrich M et al. What do people fear about cancer? A systematic review and meta-synthesis of cancer fears in the general population. Psycho-Oncol 2016; 26 (8): 1070–1079.
- University of York. People Fearful of Taking Part in Vital Clinical Research. ScienceDaily, 2020. https://www.sciencedaily.com/releases/2020/03/ 200316104010.htm. Accessed on 9<sup>th</sup> October 2020.
- Quinn GP, Koskan A, Well KJ et al. Cancer patients' fear related to clinical trial participation: a qualitative study. J Cancer Educ 2012; 27 (2): 257–262.
- Schaefer KM, Ladd E, Gergits MA, Gyauch L. Backing and forthing: the process of decision making by women considering participation in a breast cancer prevention trial. Oncol Nurs Forum 2001; 28 (4): 703– 709.
- Ghebre RG, Jones LA, Wenzel JA, Martin MY, Durant RW, Ford JG. Stateof-the-science of patient navigation as a strategy for enhancing minority clinical trial accrual. Cancer 2014; 120 (7): 1122–1130.
- Evans K, Lewis M, Hudson S (2012) The role of health literacy on African American and Hispanic/Latino perspectives on cancer clinical trials. J Cancer Educ 2012; 27 (2): 299–305.
- Byrne MM, Tannenbaum SL, Glück S, Hurley J, Antoni M. Participation in cancer clinical trials: why are patients not participating? Med Decis Making 2014; 34 (1): 116–126.
- Ford JG, Howerton MW, Lai GY et al. (2008) Barriers to recruiting underrepresented populations to cancer clinical trials: a systematic review. Cancer 2008; 112 (2): 228–242.
- Unger JM, Gralow JR, Albain KS, Ramsey SD, Hershman DL. Patient income level and cancer clinical trial participation: a prospective survey study. Jama Oncol 2016; 2 (1): 137–139.
- Unger JM, Hershman DL, Albain KS et al. Patient income level and cancer clinical trial participation. J Clin Oncol 2013; 31 (5): 536–542.
- 29. Mills N, Donovan JL, Smith M, Jacoby A, Neal DE, Hamdy FC. Perceptions of equipoise are crucial to trial participation: a qualitative study of men in the ProtecT study. Control Clin Trials 2003; 24 (3): 272–282.
- Jenkins V, Farewell D, Batt L et al. (2010) The attitudes of 1066 patients with cancer towards participation in randomised clinical trials. Br J Cancer 2010; 103: 1801–1807.
- 31. Harrop E, Noble S, Edwards M et al. "I didn't really understand it, I just thought it'd help": exploring the motivations, understandings and experiences of patients with advanced lung cancer participating in non-placebo clinical IMP trial. Trials 2016; 17: 329.
- Jenkins V, Farewell V, Farewell D et al. Drivers and barriers to patient participation in RCTs. Br J Cancer 2013; 108: 1402–1407.
- 33. van der Biessen DA, van der Helm PG, Klein D et al. Understanding how coping strategies and quality of life maintain hope in patients deliberating phase I trial participation. Psycho-Oncol 2017; 27: 163–170.
- Chi GC. The role of hope in patients with cancer. Oncol Nurs Forum 2007; 34: 415–424.
- Dellson P, Nilsson K, Jernström H, Carlsson C. Patient's reasoning regarding the decision to participate in clinical cancer trials: an interview study. Trials 2018; 19: 528.

- Truong TH, Weeks JC, Cook F, Joffe S. Altruism among participants in cancer clinical trials. Clinic Trials J 2011; 8: 616–623.
- Catt S, Langridge C, Fallowfield L, Talbot DC, Jenkins V. Reasons given by patients for participating, or not, in Phase 1 cancer trials. Eur J Cancer 2011; 47 (10): 1490–1497.
- Kim Y, Given BA. Quality of life of family caregivers of cancer survivors across the trajectory of the illness. Cancer 2008; 112 (11): 2556–2568.
- Kent EE, Rowland JH, Northouse L et al. Caring for caregivers and patients: research and clinical priorities for informal cancer caregiving. Cancer 2016; 122 (13): 1987–1995.
- Albrecht TL, Eggly SS, Gleason ME et al. (2008) Influence of clinical communication on patients' decision making on participation in clinical trials. J Clin Oncol 2008; 26 (16): 2666–2673.
- Health Research Authority (HRA). UK Policy Framework for Health and Social Care, 2017 Research. https://www.hra.nhs.uk/media/documents/ Final\_Accessibility\_uk-policy-framework-health-social-care-research\_.pdf. Accessed on 7<sup>th</sup> November 2020.

- Guillemin M, Gillam L. Ethics, reflexivity, and "ethically important moments" in research. Qual Inq 2004; 10: 261–280.
- LeBlanc TW, Kamal AH Assessing psychological toxicity and patientreported distress as the sixth vital sign in cancer care and clinical trials. JAMA Ethic 2017; 19 (5): 460–466.
- 44. National Institute for Clinical Excellence (NICE). Guidance on Cancer Services: Improving Supportive and Palliative Care for Adult Patients with Cancer. The Manual, 2004. https://www.nice.org.uk/guidance/csg4/ resources/improving-supportive-and-palliative-care-for-adults-with-cancerpdf-773375005. Accessed on 7<sup>th</sup> November 2020.
- National Cancer Institute (NCI). Common Terminology Criteria for Adverse Events (CTCAE), v5.0, publish date: November 2017. https://ctep.cancer. gov/protocolDevelopment/electronic\_applications/docs/CTCAE\_v5\_ Quick\_Reference\_8.5x11.pdf. Accessed on 9<sup>th</sup> November 2020.
- 46. National Comprehensive Cancer Network (NCCN). Distress Thermometer and Problem List Information. NCCN, 2020. https://www.nccn.org/ patients/resources/life\_with\_cancer/pdf/nccn\_distress\_thermometer.pdf. Accessed on 9<sup>th</sup> November 2020.